

REVIEW

933

Combination Atezolizumab, Cobimetinib, and Vemurafenib as a Treatment Option in BRAF V600 Mutation–Positive Melanoma: Patient Selection and Perspectives

Michelle M Dugan ¹, Matthew C Perez¹, Lilit Karapetyan^{1,2}, Jonathan S Zager ^{1,2}

¹Department of Cutaneous Oncology, Moffitt Cancer Center, Tampa, FL, USA; ²Department of Oncologic Sciences, University of South Florida Morsani College of Medicine, Tampa, FL, USA

Correspondence: Jonathan S Zager, Chief Academic Officer, Directory of Regional Therapies, Senior Member, Department of Cutaneous Oncology, Moffitt Cancer Center, McKinley Drive Room 4123, Tampa, FL, 33612, USA, Tel +1-813-745-1085, Fax +1-813-745-5725, Email jonathan.zager@moffitt.org

Abstract: The treatment landscape for advanced and metastatic melanoma has drastically changed in recent years, with the advent of novel therapeutic options such as immune checkpoint inhibitors and targeted therapies offering remarkable efficacy and significantly improved patient outcomes compared to traditional approaches. Approximately 50% of melanomas harbor activating BRAF mutations, with over 90% resulting in BRAF V600E. Tumors treated with BRAF inhibitor monotherapy have a high rate of developing resistance within six months. Combination therapy with MEK inhibitors helped to mitigate this treatment resistance and led to improved outcomes. Due to the up-regulation of PD-1/PD-L1 receptors in tumors treated with BRAF/MEK inhibitor therapy, further studies included a third combination agent, anti-PD-1/PD-L1 inhibitors. This triple combination therapy may have superior efficacy and a manageable safety profile when compared with single or double agent therapy regimens.

Plain Language Summary: Effective treatment of advanced and metastatic melanoma can be challenging. Newer treatment methods for patients with BRAF-mutated tumors include a combination of drugs with different complementary mechanisms. These drugs include BRAF-inhibitors, MEK-inhibitors, and PD-1/PD-L1 inhibitors. When these three medications are used in combination, patients may have better response rates and survival outcomes, when compared to using just one or two of these medications together. Toxicity rates are higher with a triple-medication regimen, so careful patient selection is important to consider.

Keywords: BRAF inhibitor, MEK inhibitor, metastatic melanoma, BRAF-mutant melanoma, combination BRAF therapy, triple therapy

Introduction

The treatment landscape for advanced and metastatic melanoma has drastically changed in recent years, with the advent of novel therapeutic options such as immune checkpoint inhibitors (ICI) and targeted therapies (TT) offering remarkable efficacy and significantly improved patient outcomes compared to traditional approaches.^{1–3} Immune checkpoint inhibitors include anti-programmed death-1 (PD-1) agents such as pembrolizumab and nivolumab; anti-programmed death-ligand 1 (PD-L1) agents such as atezolizumab; and anti-cytotoxic T-lymphocyte-associated protein 4 (CTLA4) inhibitors such as ipilimumab. Targeted therapies include BRAF inhibitors such as vemurafenib, encorafenib, and dabrafenib; MEK inhibitors such as cobimetinib, binimetinib, and trametinib; and dual MEK and VEGF inhibitors such as sorafenib.

Approximately 50% of melanomas harbor activating BRAF mutations, with over 90% resulting in BRAF V600E.⁴ BRAF is a serine/threonine protein kinase which activates the mitogen-activated protein kinase (MAPK)/extracellular signal-regulated kinase (ERK)-signaling pathway, easily activated by Ras; the V600E mutation is a single nucleotide mutation at codon 600, resulting in a substitution of glutamic acid for valine.⁴ This mutation causes tumors to become resistant to down-regulation of signaling feedback pathways, which leads to constitutive activation of the kinase and proliferation of tumor cells.^{4,5}

The most common cause of treatment failure with BRAF inhibitor monotherapy is acquired resistance; which has been found to occur within 5–6 months.^{6,7} Investigators have found that the mechanism of resistance involves reactivation of the MAPK pathway, upstream of MEK.^{7,8} This MAPK reactivation predicts MEK inhibitor sensitivity, which has led to the use of combination BRAF and MEK inhibitor therapy.⁸ In patients with BRAF V600 metastatic melanoma, combined BRAF and MEK inhibition is the current standard of care.^{6,9–11}

Further studies discovered that treatment with BRAF and MEK inhibitors leads to the up-regulation of melanomaspecific T cells and PD-L1. Therefore, it is understood that these actions may sensitize tumors to PD-1 and PD-L1 targeted immunotherapy.¹¹ Subsequent studies focused on triple combination therapy with a BRAF inhibitor, a MEK inhibitor, and a PD-1 or PD-L1 inhibitor, have shown promising results.^{2,12}

Double Combination Therapy with BRAF and MEK Inhibitors

A multicenter, double-blind, Phase 3 randomized controlled trial by Long et al in 2015 helped to establish combination therapy with BRAF and MEK inhibition as standard of care.⁹ This study evaluated 423 patients from 113 sites and 14 countries with BRAF V600-mutant stage IIIC or stage IV melanoma. Patients were randomly assigned to receive dabrafenib and trametinib (n = 211) or dabrafenib only (n = 212). Median overall survival (OS) was 25.1 months in the combination therapy group compared to 18.7 months in the dabrafenib group (HR 0.71, 95% CI 0.55–0.92; P=0.01). OS at 1and 2 years in the combination group was 74% and 51%, respectively, compared to 68% and 42%, respectively in the dabrafenib only group. Median progression-free survival (PFS) was 11 months in the combination group and 8.8 months in the dabrafenib only group (HR: 0.67, 95% CI: 0.53–0.84; P=0.0004). Adverse event rates were similar between the groups.⁹

The coBRIM trial, a double-blind, phase 3 randomized controlled trial in 2015 also supported combination therapy with BRAF and MEK inhibition.¹³ This study evaluated 495 patients with unresectable stage IIIC-IV BRAF-mutant melanoma, who received either vemurafenib and cobimetinib (n = 247) or vemurafenib and placebo (n = 248). Median PFS was 9.9 months in the combination therapy group, compared to 6.2 months in the vemurafenib and placebo group (HR: 0.51, 95% CI: 0.39–0.68; P<0.001). Overall response rate (ORR) was 68% in the in the combination therapy group, compared to 45% in the vemurafenib and placebo group (P<0.001). OS at 9 months was 81% in the combination therapy group, compared to 73% in the vemurafenib and placebo group (HR: 0.65, 95% CI: 0.42–1; P=0.046).¹³ A follow up study of the coBRIM trial was conducted at a median follow up of 14.2 months, and found a median PFS of 12.3 months in the combination therapy group, compared to 7.2 months in the vemurafenib and placebo group (HR: 0.58, 95% CI: 0.46–0.72; P<0.0001). Median OS was 22.3 months in in the combination therapy group, compared to 7.4 months in the vemurafenib and placebo group (HR: 0.7, 95% CI: 0.55–0.9; P=0.005).⁶

A five-year outcome study in 2019 pooled the survival data from the COMBI-d and COMBI-v trials, both of which studied combination therapy with dabrafenib plus trametinib versus dabrafenib plus placebo, for previously untreated BRAF V600-mutant unresectable or metastatic stage IIIC or IV melanoma. A total of 563 patients received combination therapy with dabrafenib and trametinib, with a PFS of 21% at four years and 19% at five years. OS was 37% at four years and 34% at five years. A complete response was observed in 19% of patients, and was associated with an improved long-term survival, with OS of 71% at five years.^{14,15}

Triple Combination Therapy with BRAF, MEK, and Immune Checkpoint Inhibitors

A review by Schmitt, Dumas, and Larkin in 2022 suggested that combination therapy with BRAF/MEK inhibitors and immune checkpoint inhibitors might achieve improved outcomes compared to single-mechanism therapy, due to high rates of progression or primary resistance to single-therapy type treatment regimens. They also suggested that triple therapy with atezolizumab, cobimetinib, and vemurafenib showed superior PFS in metastatic melanoma compared to double therapy with cobimetinib and vemurafenib.¹⁶

The Keynote-022 trial in 2020 randomly assigned 120 patients from 22 sites in 7 countries with previously untreated BRAF V600-mutated advanced melanoma to receive pembrolizumab plus dabrafenib plus trametinib (n = 60), or placebo plus dabrafenib plus trametinib (n = 60).¹⁷ After 36.6 months of follow-up, this study found a median PFS of 16.9 months in the triple therapy group, compared to 10.7 months in the double therapy group (HR: 0.53, 95% CI: 0.34–0.83). At 2 years, PFS was 41% in the triple therapy group compared to 16.3% in the double therapy group. Median duration of response (DOR) was 25.1 months in the triple therapy group and 12.1 months in the double therapy group. Median OS was not reached in the triple therapy group and s26.3 months in the double therapy group (HR: 0.64, 95% CI: 0.38–1.06). At 2 years, OS was 63% in the triple therapy group and 51.7% in the double therapy group.¹⁷ (Table 1)

Trial	Year	Sample Size	Therapeutic Agents	Follow-up (Months)	Median PFS (Months)	2-Year PFS	OS (Months)	2-Year OS	ORR	DOR (Months)	2-year DOR
Keynote 022 ¹⁷	2020	120	P/D/T (n = 60) vs Placebo/D/T (n = 60)	36.6	16.9 vs 10.7	41% vs 16.3%	Not reached vs 26.3	63% vs 51.7%	63% vs 72%	25.1 vs 12.1	N/A
COMBI-i ¹⁸	2022	532	S/D/T (n = 267) vs Placebo/D/T (n = 265)	27.2	16.2 vs 12	44% vs 36%	Not reached vs Not reached	68% vs 62%	69% (20% CR) vs 64% (18% CR)	Not reached vs 20.7	55% vs 48%
Phase I trial by Shaikh et al ¹⁹	2022	9	P/C/V (n = 5) vs P/V (n = 4)	N/A	Not reached vs 20.7	N/A	Not reached vs 23.8	N/A	N/A	N/A	N/A
IMspire 150 ¹²	2020	514	A/C/V (n = 256) vs Placebo/C/V (n = 258)	18.9	15.1 vs 10.6	N/A	N/A	N/A	N/A	N/A	N/A
IMspire 150 follow-up by Ascierto et al ²⁰	2023	514	A/C/V (n = 256) vs Placebo/C/V (n = 258)	29. I (A/C/V) and 22.8 (Placebo/C/V)	N/A	N/A	39 vs 25.8	N/A	N/A	N/A	N/A

Table I Triple Combination Therapy Trials with BRAF, MEK, and Immune Checkpoint Inhibitors for BRAF V600-Mutated Melanoma

Abbreviations: PFS, Progression-free survival; OS, Overall survival; ORR, Overall response rate; DOR, Durable response rate; P, Pembrolizumab; D, Dabrafenib; T, Trametinib; S, Spartalizumab; C, Cobimetinib; V, Vemurafenib; A, Atezolizumab.

The COMBI-i trial in 2022, a randomized, double-blind, placebo-controlled phase 3 trial, conducted at 173 centers in 29 countries, evaluated triple combination therapy with spartalizumab, dabrafenib, and trametinib in patients with BRAF V600-mutant unresectable or metastatic melanoma.¹⁸ A total of 532 patients received either triple therapy with spartalizumab, dabrafenib, and trametinib (n = 267), or double therapy with a placebo, dabrafenib, and trametinib (n = 265). Median follow up was 27.2 months. The median PFS was 16.2 months in the triple therapy group compared to 12 months in the double therapy group. Estimated 2-year PFS rates were 44% in the triple therapy group compared to 36% in the double therapy group. Patients with PD-L1-positive tumors had a longer median PFS in both treatment arms. Median OS was not reached in either treatment arm. Estimated 2-year OS rates were 68% in the triple therapy group and 62% in the double therapy group. The ORR for triple therapy was 69%, with 20% of patients achieving a complete response (CR). The ORR for double therapy was 64%, with 18% of patients achieving CR. The median DOR was not reached in the triple therapy group and was 20.7 months in the double therapy group. Estimated 2-year DOR rates were 55% in the triple therapy group and 48% in the double therapy group.¹⁸

Similar findings were supported by a smaller phase I trial which compared triple therapy with vemurafenib, cobimetinib, and pembrolizumab (n = 5) versus double therapy with vemurafenib and pembrolizumab (n = 4) for untreated BRAF V600E/K-mutant advanced melanoma. Median PFS was not reached for patients receiving triple therapy, compared to 20.7 months with double therapy. Median OS was not reached for patients receiving triple therapy, compared to 23.8 months with double therapy.¹⁹

Therapy Specifically with Atezolizumab, Cobimetinib, and Vemurafenib

The United States Food and Drug Administration (FDA) approved the use of atezolizumab in combination with cobimetinib and vemurafenib for patients with BRAF V600-mutant unresectable or metastatic melanoma in July 2020. Critical study results leading to FDA approval were from the IMspire150 trial.

The IMspire150 trial in 2020, a randomized, double-blind, placebo-controlled, phase 3 trial, evaluated triple combination therapy with atezolizumab, cobimetinib, and vemurafenib as first-line treatment for unresectable advanced BRAF V600 mutation-positive melanoma, including 112 institutions in 20 countries between 2017 and 2018. After a median follow-up of 18.9 months, the triple therapy group (n = 256) was found to have significantly longer PFS than the control group, who received cobimetinib, vemurafenib, and placebo (n = 258) (15.1 vs 10.6 months; HR 0.78; 95% CI 0.63–0.97; P=0.025).¹² A subsequent biomarker analysis of these patients was undertaken by Robert et al, which found that the PFS benefits in the atezolizumab, cobimetinib, and vemurafenib group were most evident in patients with elevated LDH and PD-L1-negative tumors.¹⁰ A follow-up study by Ascierto et al with a median follow-up of 29.1 months for the atezolizumab, cobimetinib, and vemurafenib group and 22.8 months for the control group, found a median OS of 39 months for the triple therapy group and 25.8 months for the control group, although the difference was not statistically significant.²⁰

The TRICOTEL trial in 2023, a multicenter, open-label, single-arm Phase 2 study evaluated the triple therapy regimen atezolizumab, cobimetinib, and vemurafenib in patients with previously untreated BRAF V600-mutant melanoma with brain metastases at least 5 mm (n = 65). Median follow up time was 9.7 months. Intracranial ORR was 42%.²¹

A phase Ib study in 2019 by Sullivan et al treated patients with BRAF V600-mutant metastatic melanoma with either atezolizumab and vemurafenib or cobimetinib and vemurafenib for 28 days, and then triple therapy with atezolizumab, cobimetinib, and vemurafenib (n = 48). The ORR was 71.8% and median DOR was 17.4 months, with ongoing response in 39.3% of patients after 29.9 months of follow-up.¹¹

A systematic review in 2022 by Corrie et al, which analyzed multiple treatment regimens for BRAF-mutant melanoma, found that combination therapy was superior to monotherapy with respect to OS, PFS, and ORR. Furthermore, this study found that combination therapy with atezolizumab, cobimetinib, and vemurafenib had similar results to combination therapy with encorafenib and binimetinib, with respect to OS and PFS. The ORR for atezolizumab, cobimetinib, and vemurafenib was superior to double-regimen therapies, with the exception of encorafenib and binimetinib.²² The encorafenib and binimetinib regimen was also found to have fewer treatment-related adverse events (TRAEs) compared to triple therapy with atezolizumab, cobimetinib, and vemurafenib.²²

Toxicity

Triple-therapy combination regimens tend to have a substantial but manageable toxicity.^{11,12,20} The Keynote-022 trial found a higher rate of grade 3–5 TRAEs in patients who received triple therapy (58% with pembrolizumab, dabrafenib, and trametinib) compared to those who received double therapy (25% with dabrafenib and trametinib).¹⁷ The COMBI-i trial had similar findings, with 70% of the triple therapy group and 57% of the double therapy group experiencing at least one grade 3 or higher TRAE; discontinuation of treatment due to TRAEs was 19% in the triple therapy group and 9% in the double therapy group.¹⁸ A small phase I trial which treated five patients with triple therapy (vemurafenib, cobimetinib, and pembrolizumab) was closed early due to significant adverse events.¹⁹ The TRICOTEL trial found that 68% of patients who received atezolizumab, cobimetinib, and vemurafenib experienced grade 3 or higher TRAEs.²¹ Adverse effects of the atezolizumab, cobimetinib, and vemurafenib combination regimen include increased lipase, increased blood creatine phosphokinase, hepatotoxicity, nausea, fatigue, musculoskeletal pain, diarrhea, pyrexia, and rash.^{12,20,21} Overall, the increased risk of toxicity with a triple therapy regimen must be weighed against the superior progression and survival outcomes of triple therapy compared to single or double agent therapy.

Discussion

Based on current literature, triple combination therapy with BRAF, MEK, and anti-PD-1/PD-L1 inhibition is an effective treatment regimen for patients with advanced or unresectable metastatic BRAF V600-mutant melanoma. Nearly all studies have shown superior short and long-term outcomes with this triple therapy regimen compared to monotherapy or double therapy regimens with BRAF and MEK inhibitors alone. Each of the three therapy modalities complement each other with respect to mechanism of action and sensitization. Resistance to BRAF inhibitor monotherapy predicts sensitization to MEK inhibition, and combination therapy with BRAF and MEK inhibition leads to up-regulation of PD-1/PD-L1 and sensitization to PD-1 and PD-L1 inhibition.^{8,10} The specific regimen with atezolizumab, cobimetinib, and vemurafenib has a high efficacy rate and a reasonable safety profile when used as first-line therapy.

One systematic review found a similar OS and PFS, and a higher ORR in patients treated with encorafenib and binimetinib, compared to those treated with atezolizumab, cobimetinib, and vemurafenib.²² These regimens should be compared with head-to-head randomized controlled trials to validate these results.

Various immunotherapy regimens may be considered in combination with targeted BRAF and MEK inhibition. For example, a phase 2/3 global double-blind randomized trial found that combining relatlimab, a LAG-3 blocking antibody, with nivolumab for untreated metastatic or unresectable melanoma had superior PFS (10.1 vs 4.6 months; HR 0.75; 95% CI 0.62–0.92; P=0.006) when compared to therapy with single-agent nivolumab.²³

Patient Selection

Data from randomized controlled trials support the use of upfront triple therapy with BRAF, MEK, and PD-1/PD-L1 inhibition for patients at high risk for rapid progression or death, such as those with high tumor burden at multiple distant metastatic sites.² The increased risk of toxicity with triple therapy compared to double agent therapy must be considered to maximize disease control while maintaining treatment tolerability. Ideal patient selection for triple therapy would include an acceptable baseline activity status, relatively low-risk medical comorbidities, and high disease burden. However, even in the setting of a patient with high-risk comorbidities, if the disease burden is high with multiple distant metastatic sites, triple therapy should still be considered to attempt to achieve rapid disease control. In these patients, dose modifications can be especially useful to improve tolerability.

Conclusion

Patients with advanced unresectable or metastatic BRAF V600-mutant melanoma at high risk of rapid disease progression, such as those with a high tumor burden, may benefit from upfront triple combination therapy with BRAF, MEK, and PD-1/PD-L1 inhibition. Triple therapy regimens are associated with higher toxicity, but offer superior short and long-term outcomes including OS, PFS, ORR, and DOR, when compared to single or double agent therapy. The specific regimen with atezolizumab, cobimetinib, and vemurafenib has a high efficacy rate and a reasonable safety profile.

Abbreviations

ICI, immune checkpoint inhibitors; TT, targeted therapies; PD-1, programmed death-1; PD-L1, programmed death-ligand 1; CTLA4, cytotoxic T-lymphocyte-associated protein 4; MAPK, mitogen-activated protein kinase; ERK, extracellular signal-regulated kinase; OS, overall survival; PFS, progression-free survival; ORR, overall response rate; DOR, duration of response; TRAEs, treatment-related adverse events.

Data Sharing Statement

All data supporting the statements in this manuscript have been referenced and can be found by utilizing any search engine to obtain published articles in peer-reviewed medical journals.

Ethics Approval and Informed Consent

This study, being a review article, is exempt from IRB approval and informed consent.

Consent for Publication

No previously published images or other media are being presented in this manuscript.

Author Contributions

All authors made a significant contribution to the work reported, whether that is in the conception, study design, execution, acquisition of data, analysis and interpretation, or in all these areas; took part in drafting, revising or critically reviewing the article; gave final approval of the version to be published; have agreed on the journal to which the article has been submitted; and agree to be accountable for all aspects of the work.

Funding

No funding sources specific to this paper.

Disclosure

Dr. Michelle M. Dugan, Dr. Matthew C. Perez, and Dr. Lilit Karapetyan declare no commercial or financial relationships that could be construed as a potential conflict of interest. Dr. Jonathan S. Zager has received honoraria from the Advisory Board of Delcath Systems. He has consulting agreements with Philogen, Merrit Medical, Castle Biosciences, and Merck. His institutional department receives research funding from Philogen, SWOG, Delcath Systems, and Provectus. The authors report no other conflicts of interest in this work.

References

- Dummer R, Welti M, Ramelyte E. The role of triple therapy and therapy sequence in treatment of BRAF-mutant metastatic melanoma. Response to overall survival with first-line atezolizumab in combination with vemurafenib and cobimetinib in BRAFV600 mutation-positive advanced melanoma (IMspire150): second interim analysis of a multicentre, randomised, phase 3 study. J Transl Med. 2023;21(1):529. doi:10.1186/ s12967-023-04391-1
- 2. Welti M, Dimitriou F, Gutzmer R, Dummer R. Triple Combination of immune checkpoint inhibitors and BRAF/MEK Inhibitors in BRAFV600 melanoma: current status and future perspectives. *Cancers*. 2022;14(22):5489. doi:10.3390/cancers14225489
- 3. Wolchok JD, Chiarion-Sileni V, Gonzalez R, et al. Long-term outcomes with nivolumab plus ipilimumab or nivolumab alone versus ipilimumab in patients with advanced melanoma. *J Clin Oncol*. 2022;40(2):127–137. doi:10.1200/JCO.21.02229

4. Ascierto PA, Kirkwood JM, Grob JJ, et al. The role of BRAF V600 mutation in melanoma. J Transl Med. 2012;10:85.

- 5. Pratilas CA, Taylor BS, Ye Q, et al. (V600E)BRAF is associated with disabled feedback inhibition of RAF-MEK signaling and elevated transcriptional output of the pathway. *Proc Natl Acad Sci U S A*. 2009;106(11):4519–4524. doi:10.1073/pnas.0900780106
- 6. Ascierto PA, McArthur GA, Dreno B, et al. Cobimetinib combined with vemurafenib in advanced BRAF(V600)-mutant melanoma (coBRIM): updated efficacy results from a randomised, double-blind, phase 3 trial. *Lancet Oncol.* 2016;17(9):1248–1260. doi:10.1016/S1470-2045(16)30122-X
- Hauschild A, Grob JJ, Demidov LV, et al. Dabrafenib in BRAF-mutated metastatic melanoma: a multicentre, open-label, phase 3 randomised controlled trial. *Lancet.* 2012;380(9839):358–365. doi:10.1016/S0140-6736(12)60868-X
- 8. Nazarian R, Shi H, Wang Q, et al. Melanomas acquire resistance to B-RAF(V600E) inhibition by RTK or N-RAS upregulation. *Nature*. 2010;468 (7326):973–977. doi:10.1038/nature09626
- 9. Long GV, Stroyakovskiy D, Gogas H, et al. Dabrafenib and trametinib versus dabrafenib and placebo for Val600 BRAF-mutant melanoma: a multicentre, double-blind, phase 3 randomised controlled trial. *Lancet*. 2015;386(9992):444–451. doi:10.1016/S0140-6736(15)60898-4

- 10. Robert C, Lewis KD, Gutzmer R, et al. Biomarkers of treatment benefit with atezolizumab plus vemurafenib plus cobimetinib in BRAF(V600) mutation-positive melanoma. *Ann Oncol.* 2022;33(5):544–555. doi:10.1016/j.annonc.2022.01.076
- 11. Sullivan RJ, Hamid O, Gonzalez R, et al. Atezolizumab plus cobimetinib and vemurafenib in BRAF-mutated melanoma patients. *Nat Med.* 2019;25 (6):929–935. doi:10.1038/s41591-019-0474-7
- Gutzmer R, Stroyakovskiy D, Gogas H, et al. Atezolizumab, vemurafenib, and cobimetinib as first-line treatment for unresectable advanced BRAF (V600) mutation-positive melanoma (IMspire150): primary analysis of the randomised, double-blind, placebo-controlled, phase 3 trial. *Lancet*. 2020;395(10240):1835–1844. doi:10.1016/S0140-6736(20)30934-X
- 13. Grimaldi AM, Simeone E, Ascierto PA. Vemurafenib plus cobimetinib in the treatment of mutated metastatic melanoma: the CoBRIM trial. *Melanoma Manag.* 2015;2(3):209-215. doi:10.2217/mmt.15.22
- 14. Robert C, Grob JJ, Stroyakovskiy D, et al. Five-year outcomes with dabrafenib plus trametinib in metastatic melanoma. *N Engl J Med.* 2019;381 (7):626–636. doi:10.1056/NEJMoa1904059
- 15. Grob JJ, Amonkar MM, Karaszewska B, et al. Comparison of dabrafenib and trametinib combination therapy with vemurafenib monotherapy on health-related quality of life in patients with unresectable or metastatic cutaneous BRAF Val600-mutation-positive melanoma (COMBI-v): results of a phase 3, open-label, randomised trial. *Lancet Oncol.* 2015;16(13):1389–1398. doi:10.1016/S1470-2045(15)00087-X
- 16. Schmitt AM, Dumas L, Larkin J. Atezolizumab, cobimetinib, and vemurafenib as first-line treatment for unresectable metastatic BRAF V600 mutated melanoma. *Expert Rev Anticancer Ther*. 2022;22(1):17–25. doi:10.1080/14737140.2022.2017286
- 17. Ferrucci PF, Di Giacomo AM, Del Vecchio M, et al. KEYNOTE-022 part 3: a randomized, double-blind, phase 2 study of pembrolizumab, dabrafenib, and trametinib in BRAF -mutant melanoma. *J Immunother Cancer*. 2020;8(2):e001806. doi:10.1136/jitc-2020-001806
- 18. Dummer R, Long GV, Robert C, et al. Randomized Phase III trial evaluating spartalizumab plus dabrafenib and trametinib for BRAF V600-mutant unresectable or metastatic melanoma. *J Clin Oncol*. 2022;40(13):1428–1438. doi:10.1200/JCO.21.01601
- 19. Shaikh SS, Zang Y, Hanmer J, et al. Phase I trial of pembrolizumab plus vemurafenib and cobimetinib in patients with metastatic melanoma. *Front* Oncol. 2022;12:1022496. doi:10.3389/fonc.2022.1022496
- 20. Ascierto PA, Stroyakovskiy D, Gogas H, et al. Overall survival with first-line atezolizumab in combination with vemurafenib and cobimetinib in BRAF(V600) mutation-positive advanced melanoma (IMspire150): second interim analysis of a multicentre, randomised, phase 3 study. *Lancet Oncol.* 2023;24(1):33–44. doi:10.1016/S1470-2045(22)00687-8
- 21. Dummer R, Queirolo P, Gerard Duhard P, et al. Atezolizumab, vemurafenib, and cobimetinib in patients with melanoma with CNS metastases (TRICOTEL): a multicentre, open-label, single-arm, phase 2 study. *Lancet Oncol.* 2023;24(12):e461–e471.
- 22. Corrie P, Meyer N, Berardi R, et al. Comparative efficacy and safety of targeted therapies for BRAF-mutant unresectable or metastatic melanoma: results from a systematic literature review and a network meta-analysis. *Cancer Treat Rev.* 2022;110:102463. doi:10.1016/j.ctrv.2022.102463
- 23. Tawbi HA, Schadendorf D, Lipson EJ, et al. Relatlimab and nivolumab versus nivolumab in untreated advanced melanoma. N Engl J Med. 2022;386(1):24–34. doi:10.1056/NEJMoa2109970

Cancer Management and Research

Dovepress

DovePress

Publish your work in this journal

Cancer Management and Research is an international, peer-reviewed open access journal focusing on cancer research and the optimal use of preventative and integrated treatment interventions to achieve improved outcomes, enhanced survival and quality of life for the cancer patient. The manuscript management system is completely online and includes a very quick and fair peer-review system, which is all easy to use. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/cancer-management-and-research-journal

f У in 🗖